Author's response to reviews

Title: Four minutes for a patient, twenty seconds for a relative - an observational study at a university hospital

Authors:

Gerhild Becker (gerhild.becker@uniklinik-freiburg.de)
Dorothee E Kempf (dorothee.kempf@uniklinik-freiburg.de)
Carola J Xander (carola.xander@uniklinik-freiburg.de)
Felix Momm (felix.momm@uniklinik-freiburg.de)
Manfred Olschewski (mol@imbi.uni-freiburg.de)
Hubert E Blum (hubert.blum@uniklinik-freiburg.de)

Version: 5 Date: 26 March 2010

Author's response to reviews:

Dear editors,

please find our revised final manuscript attached.

All major and minor revisions have been done. All authors read and approved the manuscript as requested. In addition, we have checked our manuscript for any typographical errors and we have made some final corrections.

We hope that our manuscript will now be well adapted for publication.

Thank you very much for your kind support.